Author's response to reviews

Title: High efficacy in Helicobacter pylori retreatment: seven days with furazolidone, levofloxacin and lansoprazole, in Sao Paulo, Brazil.

Authors:

Jaime N Eisig (eisig@uol.com.br)
Fernando M Silva (fmarcuz@hcnet.usp.br)
Ricardo C Barbuti (rbarbuti@terra.com.br)
Tomás Navarro-Rodríguez (tomasnavarro@uol.com.br)
Peter Malfertheiner (peter.malfertheiner@med.ovgn.de)
Joaquim PP Moraes Filho (joaquim.prado@uol.com.br)
Schilioma Zaterka (zaterka@superig.com.br)

Version: 3 Date: 26 December 2008

Author's response to reviews:

According to suggestions of the reviewers, the text is now more simple and clear. We believe also that the work is more appropriate to the requisites of the BMC-Gastroenterology.

Our objective to present an option for the salvage treatment of H. pylori infection with a low-cost, high efficacy and more adherence became more evident.